UNDERSTANDING INFORMATION TECHNOLOGY CULTURE IN DIGITAL-BASED PUBLIC SERVICES

Joko Tri Nugraha^{*}, Tarmizi Achmad^{**}, Hardi Warsono^{***}, Tri Yuniningsih^{***}

* Corresponding author, Faculty of Social and Political Sciences, Universitas Diponegoro, Semarang, Indonesia; Department of Public Administration, Faculty of Social and Political Sciences, Universitas Tidar, Magelang, Indonesia Contact details: Universitas Tidar, Jl. Kapten Suparman No. 39 Potrobangsan, Magelang, Jawa Tengah 56116, Indonesia ** Faculty of Economics and Business, Universitas Diponegoro, Semarang, Indonesia

*** Department of Public Administration, Faculty of Social and Political Sciences, Universitas Diponegoro, Semarang, Indonesia



How to cite this paper: Nugraha, J. T., Achmad, T., Warsono, H., & Yuniningsih, T. (2022). Understanding information technology culture in digital-based public services. Journal of Governance & Regulation, 11(2), 62–79. https://doi.org/10.22495/jgrv11i2art6

Copyright © 2022 The Authors

This work is licensed under a Creative Commons Attribution 4.0 International License (CC BY 4.0). https://creativecommons.org/licenses/by/ 4.0/

ISSN Print: 2220-9352 ISSN Online: 2306-6784

Received: 09.11.2021 Accepted: 01.04.2022

JEL Classification: H8, M1, M2, O3 **DOI:** 10.22495/jgrv11i2art6

Abstract

The increasing enthusiasm of government agencies to implement e-government can be seen as the government's commitment to anticipating the current development of science and technology (Gupta, Singh, & Bhaskar, 2018; Wang, Wang, & Liu, 2016). The present study was designed to investigate the information technology culture of public organizations in the context of public services in the Magelang City Government, Indonesia, and identify the driving and inhibiting factors of e-government based public services in the city. A mixed method with a sequential explanatory design was employed in the study. Data were garnered through questionnaire surveys, documentation, observation, and interviews. A descriptive statistic was used for the quantitative data analysis, while for the qualitative data analysis, a thematic process was conducted. Findings from the study suggest that the information technology culture in the city's governance is mainly influenced by five factors: technocratic utopianism, anarchy, feudalism, dictatorship, and federalism. The findings are crucial as they contribute to the discovery of the root cause of the problem that the implementation of e-government based public services in Indonesia is not yet optimal.

Keywords: Digitalization, E-Government, Information Culture, Public Service

Authors' individual contribution: Conceptualization — J.T.N.; Methodology — J.T.N.; Investigation — T.A., H.W., and T.Y.; Resources — J.T.N.; Writing — Original Draft — J.T.N.; Writing — Review & Editing — T.A., H.W., and T.Y.; Funding Acquisition — T.A., H.W., and T.Y.

Declaration of conflicting interests: The Authors declare that there is no conflict of interest.

1. INTRODUCTION

Recently, the effort of developing e-government has been massively carried out by the public bureaucracy. The tendency of public bureaucracy, such as ministries, non-ministerial government agencies, provincial local governments, city and district governments, to implement e-government in the governance systems is noteworthy and interesting to study for policymakers and academicians (Brewer, Neubauer, & Geiselhart, 2006; Rashid, Othman, & Narehan, 2011). Increasing enthusiasm of government agencies in implementing e-government can be seen as a commitment of the government (both the central and regional) to anticipate changes and developments in science and technology currently occur, particularly the use of information technology to support better governance and public services (Kalsi, Kiran, & Vaidya, 2008, 2009; Kalsi & Kiran, 2013; Luna-Reyes & Gil-Garcia, 2011; Johnson, 2012).



In some developing countries, the application of e-government brings advantages to the citizens and provides benefits to the national development of a country (Evans & Yen, 2006; Rehman, Esichaikul, & Kamal, 2011; Srivastava & Teo, 2010). To support the implementation of e-government, the Indonesian government in 2003 issued guidelines for regional portal infrastructure development. These were then complemented by guidelines issued in 2004 covering: quality standards and a range of services and application e-services. In 2006, the government formed the National Board of Information and Communication Technology through Presidential Decree No. 20 of 2006. The finding also indicates that through the development of information technology advancements digital connectivity in both public and private organizations will increase (Rana, Dwivedi, Lal, Williams, & Clement, 2017; Zahid & Din, 2019; Sijabat, 2020). Since the issuance of Presidential Instruction No. 3 of 2003 concerning National Policies and Strategies of E-Government Development, and Presidential Regulation No. 95 of concerning Electronic-Based Government 2018 Systems, implementation of e-government in Indonesia began to reverberate. The implementation of e-government for the government has advantages, including being more effective and efficient, reducing the level of error in providing services to citizens, and improving the government's reputation (Sharma, 2016; Gupta et al., 2018).

studies Previous have also shown the implementation of e-government has the potential to improve organizational performance in the public sector (Al-Masaeed & Love, 2014; Lau, Aboulhoson, Lin, & Atkin, 2008; Olasina, 2014; Amegavi, Bawole, Buabeng, 2018), improve the quality of policymaking through information sharing, increase interaction with citizens, and also increase private sector interaction with government (Xu, 2012; Gupta, Dasgupta, & Gupta, 2008; Gilbert, Balestrini, & Littleboy, 2004). In addition, the implementation of e-government also helps the government apply new public management principles (Elbahnasawy, 2014). Even though Information and Communication Technology (ICT) is currently becoming a commodity much talked about, it does not mean that all are successful in practicing the e-government (Ho, 2002; Moon, 2002). In Indonesia, some local government and municipalities agencies are pushing themselves to implement e-government. However, there are also numbers of other agencies or regions with no guarantee of e-government implementation yet.

Although most areas in Indonesia are already connected to telecommunications networks, optimizing the utilization of ICT infrastructure is still an obstacle. The United Nations conducts an assessment of the implementation of the E-Government System by issuing the E-Government Development Index (EGDI)¹. The EGDI is a composite measure of three important dimensions of e-government, namely: provision of online services (Online Service Index, OSI), telecommunication (Telecommunication Infrastructure connectivity Index, TII) and human capacity (Human Capital Index, HCI).

As shown in Appendix A, the EGDI Indonesia in 2018 has an index value of 0.5258. The low EGDI value is a significant contribution from the low TII which has an index value of 0.3222. Furthermore, judging from the development of TII from 2012 to 2018, Indonesia's TII did not show a significant increase.

In addition, residents have not yet optimally utilized ICT infrastructure, especially broadband networks, which is described by the low penetration of Internet users in Indonesia. Based on the results of the survey conducted by Indonesian Internet Providers Association Service (APJII, 2017). the penetration rate of Internet usage in Indonesia is 54.68% (143.26 million Internet users out of a total 262 million population). Meanwhile, the distribution of Internet user penetration rate by region is 57.70% in Java; 54.23% in Bali and Nusa Tenggara; 47.20% in Sumatra; 72.19% in Kalimantan; 46.70 % in Sulawesi and; 41.98% in Maluku and Papua. The low penetration rate of Internet users in several parts of Indonesia is due to the low quality and limited capacity of the available broadband network.

Although in 2014 the city of Magelang has compiled a grand design for the development of telematics for the city government, most of the current e-government activities are still in the form of providing websites and information services. Based on the final report of Bappeda² (2020), several problems were found that need to be considered in the application of ICT in Magelang City, including: 1) lack of readiness from the government, both in terms of human resources and facilities; 2) technology and information infrastructure is not adequate, the wide area network (WAN) system has not been implemented optimally; 3) the information system is developed with various technology platforms, making it difficult to integrate; 4) obstacles in the pattern of electronic data management (in the form of a database), so that it will be difficult for the data entry process; 5) there is no clear bureaucratic policy regarding the authority for data ownership and access by other parties; 6) lack of understanding of the principles of IPR so that some work units use pirated software (illegal or without a license); 7) IT control in several work units is still limited to operators, not at the system analyst/programmer level, so there is still a need to improve the quality of HR in the IT field; 8) not yet optimal cooperation between Local Government Work Unit (Satuan Kerja Perangkat Daerah, SKPD) in the management and provision of data; 9) public information has not been managed properly; and 10) lack of awareness of SKPD in managing websites and information systems.

of the most influential factors on One the success of the e-government implementation is information culture, in addition to the human resources, infrastructure, and funding factors (Kalsi & Kiran, 2013; Kalsi, Kiran, & Vaidya, 2008, 2009), it is essential to study such a topic. Human resources do play a pivotal role and are a success factor of public services based on e-government because human resources are the one that drives other resources (Luna-Reyes & Gil-Garcia, 2011; Johnson, 2012). However, no matter how good the e-government infrastructure is if there is no awareness concerning information culture, all

² Magelang Municipal Development Planning Agency



¹ https://publicadministration.un.org/egovkb/en-us/About/Methodology

investments carried out will be in vain without maximum results (Khamis & van der Weide, 2017; Oni, Okunoye, & Mbarika, 2016; Pederson, 2016; Urbina & Abe, 2017).

Within this context, this paper presents the concept of information culture in the city of Magelang, Indonesia, identifies the types of information culture, discusses some of the main challenges and opportunities in providing e-services. The objectives of this paper are, to discuss the information culture challenges and key opportunities of e-government and to identify information culture which has a high strategic impact in developing countries. This research contributes to finding the root cause of the not yet optimal implementation of digital-based public services in Indonesia.

This paper is organized as follows. Section 1 presents an overview of the problems of implementing e-government in developing countries, especially the city of Magelang, Indonesia. Section 2 provides an overview of e-government and types of information culture. Section 3 contains the research methods used. Section 4 presents the results of the research on the types of information culture. Section 5 contains a brief discussion of the factors that influence information culture in developing countries as well as recommendations for city governments in Indonesia to improve the quality of digital-based services. Finally, in Section 6 the conclusions of this study are presented.

2. LITERATURE REVIEW

2.1. E-government concept

Various theoretical meanings have been put forward non-government institutions either bv or government institutions towards the concept of e-government. United Nations Development Programme (UNDP) has defined, "E-Government is the application of Information and Communication Technology (ICT) by government agencies" (Indrajit, 2002, p. 2). One of the goals of using information technology in government agencies is to improve better services that are an advantage for citizens to get better public services, which are more transparent, accountable, reduce time and costs, empower people through information and reduce the level of corruption (Borah, 2013; Saxena, 2005; Abbassy & Mesbah, 2016; Gupta, Bhaskar, & Singh, 2016a; Gupta, Singh, & Bhaskar, 2016b).

Earlier, Berryman (2004) defines e-government online as an service process, e-service. e-administration, and e-democracy. E-government is intended to provide ease of communication and transactions between the government and its citizens (Government-to-Citizen, G2C), the government with business organizations (Government-to-Business, G2B), and among government institutions (Government-to-Government, G2G; Government-to-Nonprofit, G2N) and also the government and its staff (Governmentto-Employee, G2E) (Fang, 2002; Fang & Wang, 2006; Basu, 2004; Javidian & Mollayaaghobi, 2011;Ganapati & Reddick, 2012).

If these four concepts are concluded, then e-government is a government system process using ICT as a tool to provide ease of communication and transaction processes to citizens, business organizations, and among government agencies and their staff. So that efficiency, effectiveness, transparency, and accountability of government can be achieved for the people (Slamet, Hamdan, & Deraman, 2007; Al-Sarabi & Khanfar, 2010).

This e-government can be implemented in various ways including 1) providing information sources, especially information often sought after by the public; 2) provision of access mechanisms through information kiosks available at government offices and also in public places; and 3) e-procurement where the government can conduct tenders online and transparent (Raharjo, 2001).

Development efforts for e-government can be done through several stages or levels. One way to categorize these types of services is to look at them from two main aspects, namely: 1) aspect of complexity, which concerns how complicated the anatomy of an e-government application which is about to be built and applied and 2) aspect of benefits, which involves matters relating to the magnitude of the benefits felt by its users (Indrajit, 2002, 2005).

2.2. Information culture in the context of public organizations

Information culture is a relatively recent concept but has attracted the interest of researchers from around the world. The term information culture was first introduced by a researcher from Finland, Mariam Ginman (Oliver, 2017). Several decades ago, Ginman (1987) contend that the concept of information culture as one that is open to change and innovation, where management is responsive to information about the external environment. In other words, information culture is presented as a necessary and desirable state if the organization is to be successful and competitive. Ginman's research becomes the basis for research by other researchers (Hansen & Widén, 2016; Widén-Wulff, 2000; Widén & Hansen, 2012). However, this perspective on information culture has been broadened by other researchers (Oliver & Foscarini, 2014) to include all cultural settings, i.e., where information can be poorly managed as well as very positive situations processes systems and where have been implemented successfully and effectively used appropriately by all stakeholders.

To summarise, there are two groups of definitions of information culture reflected in the literature (Oliver, 2017). One cluster considers information culture to be "information culture", that is, an environment in which information is well managed and used effectively and efficiently. Another cluster considers the concept to reflect cultural values, attitudes, and behaviors related to information, whether it is positive or negative, effective or ineffective. It is this latest, more holistic, and comprehensive articulation of an information culture that is the subject of this paper. In this way, all organizations and communities, no matter where they are in the world, will have an information culture. Furthermore, the more complex the entity, the more likely it is to be characterized by multiple information cultures. In archival storage (i.e., activities that include the current management of records and short-term records), this conception of information culture has been recognized as an important feature that needs to be taken into account when developing and implementing a records retention strategy. Evidence of this recognition is provided by the fact that information culture is one of three aspects for analysis identified in record-keeping informatics, a new approach to records management that has been developed by Australasia authors (see Upward, Reed, Oliver, and Evans, 2013 for an outline of features recordkeeping informatics and a motivator for its development).

Sociologists define culture as interactive patterns and collective behavior that are recognized and practiced by a group of actors (Turner, 1998). In society, culture only builds social relationships and reflects how community members live their daily lives (Altman & Baruch, 1998). Such a reflection is largely based on a set of shared values that distinguish members of one group from another (Hofstede, 1981). These shared values help members of the same group deal with their social relationships with whom I am and how I should behave. Actors with the same cultural values can recognize each other and build their social identity in a group (Thompson & Wildavsky, 1986). In an organization, the collective behavior of actors and social identity will shape the overall organizational culture.

Theoretically, organizational culture as a pattern of shared basic assumptions learned by groups in an organization as a tool to solve problems with adjusting external factors and integration of internal factors and has been proven valid and is therefore taught to new organizational members as a correct way to perceive, think, and feel in relation to the problems at hand (Schein, 1992). A previous study by Lubbad and Bohari (2016) found an interaction between the influence of organizational characteristics (management support, information technology readiness, government strategy) and organizational culture, so as to explain how information technology innovations emerged in an organization.

Similarly, Winarsih (2008) distinguishes the existence of four kinds of organizational culture, namely:

1) *Apathetic culture:* In this type, the attention of the organization members towards the relationship between humans and the attention to the performance in carrying out tasks are both low.

2) *Caring culture:* This type of organizational culture is characterized by low attention to performance and high attention to human relations.

3) *Exacting culture:* The main characteristic of exacting culture is that attention to people is very low, but attention to performance is very high.

4) *Integrative culture:* In organizations that have integrative culture, attention to people and attention to performance are both very high.

For such organizations, companies, and public organizations, information is everything. Not only because the information is part of the process of creating goods and services, but almost every important decision-making management needs quality information. But in reality, there are still organizations that are very rigid in applying management principles information in their possession, so it is very difficult for anyone just to access it (Indrajit, 2012). In the information technology domain, cultural studies have paid great attention to the impact of culture on business practices and technology development (Kass, 1991; Balsamo. 1996: Katz & Townsend. 2000). For example, cultural norms in Europe were found to affect all aspects of ICT strategies including product development, sales strategies, sales cycles, and product deployments (Tse, Tiong, & Kangaslahti, 2004); virtual team decision-making was found to be significantly influenced by cultural diversity and the communication process was hindered by it (Shachaf, 2008); organizational culture also plays a distinctive role in the organization's decision to adopt high-tech products (Sharma, 1994); more people-oriented elements specifically, of organizational culture such as trust, teamwork, autonomy, flexibility, and information sharing, seem to have more positive influence on successful IT implementation (Harper & Utley, 2001).

At an extreme point, there are organizations that are extremely rigid in applying management principles to the information they have, which makes it very difficult for anyone to access it. While on the other hand, there are quite a number of more moderate organizations, where these organizations are free and open, for anyone and anywhere who wants to obtain information for the needs of the organization's daily activities can easily obtain and access it (Chen & Bennett, 2010; Da Veiga, 2016).

Boisot (as cited in Canals, Boisot, & MacMillan, 2005, p. 5; Indrajit, 2012, p. 2; Wang et al., 2016, pp. 1-2) defines information culture as a conducive system that supports the occurrence of information exchange behavior both among individuals and groups in organizations. In his famous work, viz Boisot's Model, he stated that the structure of information management will be closely related to the characteristics of information along with the context of the organization's existence, hence, it can be categorized into two matrix coordinates as follows. 1) Codified vs. uncodified, meaning information is considered as codified if needed a categorization mechanism based on a certain standard code as like substances in chemical reactions, variables in physical formulas, rank in the military, and so forth. While uncodified information is often found in representations such as magazines, newspapers, television, radio, and so on. 2) Diffused vs. undiffused, meaning information is considered as diffused if freely accessible by the public. While categorized as undiffused if only allowed to be accessed by certain groups of individuals or communities.

Several previous studies have explained that organizational culture is an important and influential part of the success or failure of implementing an information system (Bulinska-Stangrecka, 2016; Mardiana, Tjakraatmadja, & Aprianingsih, 2018; Oliver, 2017). Based on the results of his research, which was inspired by Max Boisot's theory, Keen (as cited in Indrajit, 2012, p. 2) found that there are five types of information management structure models that are strongly influenced by the organization information culture. The five models and their characteristics are as follows:

1) Technocratic utopianism, a system in which the organization is strict, in detail, and consistently regulates the creation, distribution, and use of each category of information available in the organization. 2) Anarchy, is a condition where the organization has no policies and procedures at all related to information management.

3) Feudalism, occurs when the information management needs and governance are held or monopolized by one or several specific organizational functions. These organizational units that in fact determine the models, categories, and information standards that need to be managed by the organization and they will provide it for all individuals.

4) Dictatorship, places the position of the leader of the organization or commonly called the head of the office/work unit who decides and controls the existence of information in the organization.

5) Federalism, is seen as a fairly democratic management system because a number of interested parties hold a joint consensus regarding the management of existing information and flow in the organization.

several There are factors that cause an organization to become more competitive, including organizational structure, decision-making model, and managerial style. Based on the results of his study, the focus of organizational culture and knowledge about the organization are factors that require attention (Indrajit, 2012). In line with the development of the concept of governance, in the science of public administration, there is also the development of new concepts. In the 1980s and early 1990s, a new managerial approach emerged in the public sector as a response to the shortcomings of the traditional public administration model. This approach is called the new public management (Hughes, 1994). In the new public management, there are fundamental changes, both in terms of strategic aspects, internal management, and in terms of the relationship between the bureaucracy, politicians, and the public. Hoods (as cited by Hughes, 1994) says that in essence there are four types of changes proposed by the new public management approach, focusing namely: the organization on output, managing organizational inputs through contracts, reducing the scope of government work, and change the accountability relationship between the bureaucracy and politicians and the public. Changes in the life of the nation and state fundamentally demand clean, transparent government governance that can respond to the demands of change. Information technology provides an opportunity for structuring various aspects of the life of the nation and state which are increasingly open, where the interests of citizens have a central position. E-government provides opportunities for structuring various aspects of national and state life that are increasingly open, where the interests of citizens have a central position (Batara, Nurmandi, Warsito, & Pribadi, 2017; Hooda & Singla, 2021; Hung, Chang, & Kuo, 2013; Stemberger & Jaklic, 2007; Weerakkody, Janssen, & Dwivedi, 2011).

3. METHODOLOGY

The present study employed a mixed method with a mixed design partially dominant in sequence (Leech & Onwuegbuzie, 2007, 2009). The dominant sequential mix design is research with two phases that occur sequentially, but one of the quantitative phases has a more dominant emphasis. The survey technique used in this study employed a sequential explanatory design that occurs in two different interactive phases (Bowen, Rose, & Pilkington, 2017; Côté-Leclerc et al., 2017; Creswell & Creswell, 2018; Creswell & Piano Clark, 2018; Ivankova, Creswell, & Stick, 2006; Jenkins, 2019; Muhaimin et al., 2019). In this design, quantitative numerical data were collected and analyzed in the first phase. Meanwhile, qualitative text data were collected and analyzed in the second phase to help explain or describe the quantitative results obtained in the first phase (Bungin, 2020; Krohwinkel, 2015; Maleku, Kim, Kagotho, & Lim, 2021; McKim, 2017; Morren, Gelissen, & Vermunt, 2013). The basic procedure for applying the explanatory sequential design in the study can be seen in Table 1.

| Research design | Research approach |
|--|---|
| Theoretical paradigm | A pragmatic, explanatory, systematic approach. Quantitative research using Statistical Package for Social Sciences (SPSS). Qualitative research (interviews). |
| Methodology | Explanatory sequential approach. Questionnaire/survey, case studies (interviews). A mixed method. |
| Participants | Website manager in each regional apparatus organization (Organisasi Perangkat Daerah, OPD). |
| Data collection method | Questionnaire/survey offline. Interviews carried out; one to one with web managers in each OPD. |
| Data analysis | Questionnaire/survey analyzed and evaluated using SPSS. Thematic analysis of qualitative data gathered, analyzed using manual method. |
| Validity, reliability Measures and factors will be taken into consideration to ensure that the r valid and reliable as possible. | |

3.1. Design of the study

A mixed method study with a sequential explanatory design was conducted to explore research phenomena involving multiple data sources and establish the findings' reliability and validity (Creswell & Creswell, 2018). The sequential explanatory design relies on quantitative data supported by qualitative data to achieve the research objectives (Creswell & Creswell, 2018). Through this method, we investigated the type of information culture in the Magelang City Government. The research was conducted for eight months from November 2017 to June 2018 in the city of Magelang, Indonesia. We used two types of data sources, surveys, and interviews to answer the research questions. In mixed method design, several data methods such as surveys, observations, interviews, and document analysis are significant (Stake, 1995). Data collection methods should be



linked to the research approach. Both methods of data collection are important to provide complete and detailed information in this study. This research begins with a quantitative process followed by the collection and analysis of qualitative data.

3.2. Quantitative data

We first surveyed all website managers of regional apparatus organizations (OPD) in the city of Magelang, Indonesia. However, the sample is 30 who manage websites in each OPD. We distributed the printed survey instruments to 30 respondents of which 30 instruments were returned and all 27 question items were answered by the respondents. Based on construct validity, several words were used to capture respondents' perceptions, such as "Magelang City Government held training for website managers". The final survey of this study consisted of 27 items for information culture-type questions. The questionnaire also contains five items for respondent information (among others, gender, age, last education, years of service and agency). We distributed survey questionnaires assisted by two research assistants. After data collection, we entered the data into the computer and saved it into a Microsoft Excel file. Then we computed the data and measured frequency, percentage, mean, and standard deviation (descriptive statistics) to describe the type of information culture.

The sampling technique used in this study is a simple random sampling approach, where the researcher takes 30 sample members from the population at random without regard to the existing strata in the population from each OPD managing e-government in the city of Magelang. The data is taken from four categories of institutions within the Magelang City Government, Indonesia, including:

1) The secretariat, consisting of the regional secretariat office and the secretariat office of the Regional People's Representative Council (DPRD).

2) The agency, consisting of the Office of the Inspectorate; the Office of the Regional Planning Agency (Bappeda); the Office of the Research and Development Agency (Balitbang); the Office of the Personnel, Education and Training Agency (BKPP); the Office of the Regional Financial and Asset Management Agency (BPKAD); the Office of the Unity of the Nation and Politics (Kesbang); and Regional General Hospital (RSUD).

3) The service, consisting of the Department of Education and Culture (Dikbud): the City Health Office (DKK); the Department of Public Works and Spatial Planning (DPUPR); the Department of Housing and Settlements (Perkim); Social Service (Dinsos); the Department of Manpower (Disnaker); Office of Community Empowerment, Women, Child Protection, Population Control and Family Planning (DPMPPAPKB); the Department of Agriculture (Dipertan); Environmental Service (DLH); the Department of Population and Civil Registration (Disdukcapil); the Department of Transportation (Dishub); the Office of Communication, Information and Statistics (Diskominsta); One-Stop Integrated and Investment Service (DPMPTSP): Service the Youth, Sports and Tourism Office (Disporapar); Library Service (Library), Industry and Trade Service (Disperindag); Service of the Civil Service Police Supervision Unit (Satpol).

4) Sub-districts, consisting of North Magelang sub-district, a district of Central Magelang and South Magelang district.

3.3. Qualitative data

In order to better understand the understanding of the types of information culture in each of the Magelang City Government agencies, we interviewed eight participants who filled out "yes" in the statement of agreement in the survey. The semistructured interview questions were adapted from the survey items. These questions are listed to understand how the information culture exists in each agency and how they understand the policies of leaders in developing information technology. This method allows the interviewer to talk about issues that may not be addressed in other study data collection methods (Creswell & Creswell, 2018). Interviews were conducted to obtain in-depth information on the integration of technology in science subjects. The interviews were conducted in Indonesian and lasted from 50 to 60 minutes. masked participants' identities We using pseudonyms for ethical considerations to safeguard human rights protection. For data analysis, we applied case analysis (Miles & Huberman, 1994). We interviewed eight participants individually using interview notes. We analyzed and categorized transcripts of interview notes into thematic items. This process was repeated until the last participant, eight participants. We marked the relevant statement pieces, putting the relevant statement pieces into the permanent category (Creswell & Creswell, 2018). For trustworthiness (Miles & Huberman, 1994) from this research, we discussed verbatim statements of interview transcripts followed by member examination procedures (Creswell & Creswell, 2018). We carried out the screening procedure with all interview participants and all research members. We provided interview data to participants for their feedback and consent. This step was taken to reduce research bias. Participants agreed on the research data to be presented.

4. FINDINGS

4.1. Dimension of technocratic utopianism

The dimension of technocratic utopianism consists of 8 indicators. In the first statement (see Appendix C, Table C.1), 15 (50.0%) respondents agreed that technology-based services are more effective and efficient. While the remaining 14 (46.6%) respondents strongly agree with the statement. The number represents that half of the respondents agree on the efficiency and effectiveness of technology-based services.

In the second statement (see Appendix C, Table C.2), 20 (66.6%) respondents agree that the Magelang City Government has formulated regulations on digital-based services. Meanwhile, 8 (26.6%) respondents mention that the Magelang City Government has provided regulation on digital-based services, and 2 (6.6%) respondents giving no opinion.

The third statement (see Appendix C, Table C.3) mentions that the Magelang City Government held website training for administrators attains agreement from 15 (50.0%) respondents. Twelve (40.0%) respondents said that the Magelang City Government held training for website administrators. While the remaining 3 (10.0%) respondents gave no opinion. Thus, half of the respondents agreed that the Magelang City Government held training for administrators of the website.

The fourth statement Table C.4) specifying that Appendix C, (see administrators of the website in OPD have ICT competency gathered respondents. from (53.3%) agreement 16 Simultaneously, 12 (40.0%) respondents expressed their strong agreement with the statement, while the remaining 2 (6.6%) respondents gave no opinion. Thus, more than half of the respondents agreed that the administrators of the website in OPD have ICT competency.

The fifth statement (see Appendix C, Table C.5) saying that the Magelang City Government is allocating funds for e-government application development obtains strong agreement and concession from 10 (33.3%) and 18 (60.0%) respondents, while 2 (6.6%) respondents gave no opinion.

Similarly, 5 (16.6 %) and 22 (73.3 %) respondents strongly agree and agree, respectively, with the sixth statement (see Appendix C, Table C.6) explaining that Magelang City Government held cooperation with third parties in developing e-government services. While the remaining 3 (10.0%) respondents gave no opinion.

The seventh statement (see Appendix C, Table C.7) stating that the Magelang City Government has built the infrastructure for the development of e-government applications has gathered strong agreement and agreement from 9 (30.0%) and 20 (66.6%) respondents, respectively. The eighth statement (see Appendix C, Table C.8) that states the respondent's nescience on the benefits of e-government based public services attains disagreement from 20 (66.6%) respondents and strong disagreement from 8 (26.6%) respondents. While the remaining 2 (6.6%) of respondents gave no opinion. Thus, almost all of the dimensions of technocratic utopianism have attained agreement from more than half of the respondents.

From the frequency distribution data above, every individual in the organization must have information technology literacy which is a valuable asset that is inseparable from the existence of the organization. The following is an excerpt from an interview with one of the informants.

"Does the Magelang City Government regularly hold information technology training?", the informant (Ang) was asked. The answer was as follows:

"The Magelang City Government regularly conducts IT training, but our difficulty is that there is still limited staff with an IT education background. Not all employees are able to operate information technology. So far, the Magelang municipal government has also collaborated with third parties in providing information technology-based services".

4.2. Dimension of anarchy

In order to perceive the anarchy dimension side, this study uses 5 indicators. The first statement of the anarchy dimension (see Appendix D, Table D.1)

is leaders establish a special policy regarding information management. Twenty-one (70.0%) respondents agree with this statement, while 6 (20.0%) respondents strongly agree with the statement. The remaining 3 (10.0%) respondents gave no opinion.

The second statement (see Appendix D, Table D.2) is the presence of an standard operation procedure (SOP) on information management in every OPD. Frequent distribution of SOP concerning information management in every OPD gathered agreement from 22 (73.3%) respondents and strong agreement from 6 (20.0%) of respondents, with the remaining 2 (6.6%) of respondents giving no opinion.

The third statement (see Appendix D, Table D.3) states that the Magelang City Government enables the public to access information via website indicates 18 (60.0%) respondents agree that the office open access to public information via the website, while 11 (36.6%) respondents strongly agree that the office opening access to public information via the website. Further, 14 (46.6%) and 3 (10.0%) respondents agree and strongly agree with the fourth statement (see Appendix D, Table D.4) mentioning that the office should involve outsourcing manpower in website management.

The last statement (see Appendix D, Table D.5) states that not all vital information is accessible to the public. Eighteen (60.0%) and 4 (13.3%) respondents agree and strongly agree to the statement. Meanwhile, 4 (13.3%) respondents disagree that not all vital information is accessible to the public, and the remaining 1 (3.3%) respondent strongly disagreed with the statement. Therefore, more than half of the respondents agree with all of the indicators of the anarchy dimension.

From the data above, in this dimension, there is individual freedom to access information in accordance with the main tasks of their respective functions. The following is an excerpt from an interview with one of the informants.

"Are there any rules or SOPs for managing information in each OPD?", the informant (Yn) was asked. The answer was as follows:

"Regarding the rules or SOPs for managing information systems, they actually already exist, however, their implementation is still constrained by infrastructure limitations so that between OPDs cannot be maximized in sharing information".

4.3. Dimension of feudalism

In the cultural side of feudalism, this study uses 6 indicators. The first statement (see Appendix E, Table E.1) illustrating the substantial importance of implementing digital technology in public services obtains strong agreement from 13 (43.3%) respondents and agreement from 16 (53.3%) respondents. Thus, more than half of the respondents agreed that implementing digital technology in public services is extremely important.

The second statement (see Appendix E, Table E.2) stating that the website managers in respective OPD must comprehend information technology gains strong agreement from 14 (46.6%) respondents and agreement from 15 (50.0%) respondents. The third statement (see Appendix E, Table E.3) states that digital technology will facilitate task completion, which is agreed by 16 (53.3%) respondents. Meanwhile, 12 (40.0%) respondents express their strong agreement with this statement.

The fourth statement (see Appendix E. Table E.4) explains that web managers in every OPD had an education background is agreed by 16 (53.3%) respondents and strongly agreed by 8 (26.6%) respondents. Meanwhile, 2 (6.9%) respondents gave no opinion and the remaining 2 (6.6%) respondents disagree that web managers in every OPD had education in information technology.

As for the fifth statement (see Appendix E, Table E.5) that every employee of the Magelang City Government is capable of operating computers indicates, attains strong agreement from 5 (16.6%) respondents and agreement from 13 (43.3%) of respondents. Whereas the remaining 6 (20.0%) gave no opinion.

The sixth statement (see Appendix E, Table E.6) choosing to do tasks manually instead of using a computer obtain agreement from 17 (56.6%) respondents and strong agreement from 7 (23.3%) respondents. Meanwhile, 4 (13.3%) respondents gave no opinion and 1 (3.3%) respondent disagreed that choosing to do tasks manually instead of using a computer. Thus, more than half of the respondents agree on every indicator of the feudalism dimension.

From the above, in this dimension, the need and governance of information management are held by one or more specific organizational functions. The following is an excerpt from an interview with one of the informants.

"Because the governance of information management is held by one or several specialized organizations, what is the level of e-literacy of the website managers in each OPD?", the informant (Dn) was asked. The answer was as follows:

"In general, the literacy level of website managers is good, however, we are still constrained hν inadequate information and technology infrastructure, the WAN system has not been implemented optimally and mastery of information technology in several work units is still limited to the operator level. Not yet at the system analyst/programmer level, so there is still a need to improve the quality of human resources (HR) in the field of information technology".

4.4. Dimension of dictatorship

There are 4 indicators used to investigate the cultural side of dictatorship. The first indicator (see Appendix F, Table F.1) states that all information in OPD is passed through the chairperson first. It gains strong agreement from 7 (23.3%) respondents and agreement from 17 (56.6%) respondents. Three (10.0%) respondents give no opinion and the remaining 3 (10.0%) respondents disagree that every information in OPD is passed through the chairperson first.

The second statement (see Appendix F, Table F.2) explains that OPD leaders decide which information is accessible to the public attain strong agreement from 9 (30.0%) respondents and agreement from 18 (60.0%) respondents. Two (6.6%) respondents gave no opinion and another 1 (3.3%) respondent disagree that OPD leaders decide which information is accessible to the public.

The third statement (see Appendix F, Table F.3) states the presence of a special unit (division) of information management in every OPD has been agreed by 19 (63.3%) respondents and strongly agreed by 6 (20.0%) respondents. Three (10.0%) respondents give no opinion, while 2 (6.6%) respondents disagree with the statement.

The fourth statement (see Appendix F, Table F.4) states that leaders create the flow of information distribution system. This indicator has been agreed by 22 (73.3%) respondents and strongly agreed by 6 (20.0%) respondents, while 2 (6.6%) respondents gave no opinion. Thus, more than half of the respondents agree with all indicators of dictatorship cultural side.

In this dictatorship dimension, it is usually the head of the office or the regional work unit³ who decides and controls the existence of information in the organization. The following is an excerpt from an interview with one of the informants.

"Since the head of the office decides the existence of the information, what is the structure of information access control in each OPD?", the informant (Wln) was asked. The answer was as follows.

"Formally in each OPD, there is an office head or satker who will determine the policy whether the information is accessible to the public or not. In addition, at our OPD there are obstacles in the pattern of electronic data management (which is still in the form of a database) so it will be difficult for the data entry process. Meanwhile, another obstacle is the absence of a clear bureaucratic policy (SOP) regarding the authority for data ownership and access by other parties that support the disclosure of public information".

4.5. Dimension of federalism

To examine the cultural side of federalism, this study uses 4 indicators. The first statement (see Appendix G, Table G.1) illustrates that the leaders open the flow of information accessible to all employees obtain agreement from 19 (63.3%) respondents and strong agreement from 6 (20.0%) respondents. Two (6.6%) respondents gave no opinion and 3 (10.0%) respondents disagree that leaders open the flow of information accessible to all employees.

The second statement (see Appendix G, Table G.2) states that leaders always share information and discuss it with all employees. Twenty-one (70.0%) respondents agree and 6 (20.0%) respondents strongly agree with this statement. Three (10.0%) respondents gave no opinion. The third statement (see Appendix G, Table G.3) that mentions that leaders should hold discussions regarding information management with employees attain agreement from 24 (80.0%) respondents and strong agreement from 6 (20.0%) respondents.

The fourth statement (see Appendix G, Table G.4) states the presence of a special unit of information managers in every OPD gather agreement from 22 (73.3%) respondents and strong agreement from 6 (20.0%) respondents. One (3.3%) respondent gives no opinion and another 1 (3.3%) respondent disagrees with the presence of a special unit of information managers in every OPD. Thus, more than half of the respondents agree with all indicators of the federalism cultural side.

³ Satuan Kerja Perangkat Daerah (Satker).



In this dimension, there is a consensus about the governance of information that exists and flows in the organization. The following is an excerpt from an interview with one of the informants.

"Is there a special information management unit in each OPD?", the informant (Brt) was asked. The answer was as follows.

"In every OPD there is already an information management unit. However, currently, it is still constrained by the condition of broadband infrastructure in Magelang City. Currently, the work units in the mayor's office complex are connected to a local network (LAN) using fiberoptic cables. Meanwhile, the work units outside the mayor's office complex are not yet connected to the intranet network. For WAN connections, most OPDs go through Internet connection lines by subscribing independently. With a WAN configuration like this, it is relatively difficult to control ICT activities in each OPD, because each OPD is very independent to manage its own network. Each OPD unit is directly connected to the international Internet network. The existing network is that every OPD is free to interact with the Internet so that the security of data and information transactions is not maintained, so it is very vulnerable to interference because every point is in an open area that is directly dealing with the public".

5. DISCUSSION

5.1. Inhibiting factors in building an information culture in the Magelang City Government

Generally. government organizations around the world present massive investments in the development of e-government. Meanwhile, the failure in implementing e-government is mainly caused by human resources (Nugraha, 2018; Sheth & Sarma, 2007). Our findings in several OPD units of the Magelang City Government are still short of with information employees technology backgrounds so that the existing employees are still only at the operator level, not at the system analyst/programmer level, so there is still a need to improve the quality of human resources by recruiting new employees.

Human resource factors, especially those related to literacy and culture, remain to be an inhibiting factor in the Magelang City Government e-government application. E-literacy is defined as the ability to use information technology, including the ability to use the Internet (Internet literacy). Meanwhile, the IT culture has not been extensively developed within the Magelang City Government itself and in the wider community. Consequently, Magelang City Government officials, as service providers, tend to use offline services, because they prefer traditional direct services, such face-to-face, as correspondence, and using the telephone. Sheth and Sarma (2007) identify challenges in the implementation of e-governmentbased public services, including: 1) minimum ICT advancement in developing countries; 2) Internet fraud, estimated to cost USD 2.8 billion; 3) privacy issues; 4) intrusive services (such as cell-phonebased) since customers may not like to be contacted by phone at any time and any place.

The first and primary challenge of e-government applications is the low Internet development, primarily in the city of Magelang where Internet access and speed have not been expanded equally due to limited funds for infrastructure development. Besides, people's reluctance to use the Internet is correlated to fraud in transactions and Internet business. Meanwhile, the growing concern on privacy issues obligates all information, transactions, and communications between government agencies and the public to follow certain rules. The government must also ensure the data privacy of each person while guaranteeing that they provide the proper and trustworthy information. The presence of spyware and security holes in operating systems has expanded people's idea that Internet transaction is equal to compromising their privacy. Therefore, it is necessary to have a security, information, and privacy act to implement an e-government system. Lastly, e-services demand people choose the time wisely, as some people are not happy to be contacted at a specific time (Evanschitzky & Iyer, 2007).

Furthermore, Rahardjo (2001)discover the obstacles and challenges of e-government implementation in the city of Magelang. The culture of sharing does not yet exist. The culture of sharing is the sharing of information to ease human activities. It has not been entrenched in Indonesia, even in the public bureaucratic service there is the adage "If you can make it difficult ... why make it easier". Many people take the opportunity to make it difficult to get this information. One more of such obstacles and challenges is the culture of documenting bureaucratic apparatus that is not commonly practiced. One of our biggest difficulties is the lack of documentation capabilities following ISO 9000 and software engineering standards. Another one is the scarcity of reliable human resources. As information technology is a new field, the government lacks reliable personnel in the ICT field. Most of the reliable human resources usually choose private business and industry, hindering the implementation of e-government. Further, this government's lack of capacity is often exploited by industrial companies to sell wrong and expensive solutions. Inadequate and expensive infrastructure has caused inegual advancement of telecommunications infrastructure in Indonesia. Many areas in Indonesia have no telephone lines or even electricity lines, while some of them have expensive access to the telecommunication infrastructure. Unlike, other countries governments and communities work together to create many affordable access points, for example, in public libraries, the Indonesian government only provides minimum access points. As for the city of Magelang, the access point can be provided at the post office, government office, and other public places.

E-government realizes transparency and accountability of public services, increases efficiency, productivity, and effectiveness of government administration, and can improve the quality of services to the community. However, the early-stage implementation of e-government in the city of Magelang encounters a number of challenges, such as limited extent and Internet network, hardware, and a low number of human resources with great ICT mastery.



5.2. Policy recommendations for the Magelang City Government

The need for reform in the public sector should be point for a pressing bureaucratic reform. The bureaucratic reformation process is not an easy job. It takes a clear and well-directed concept in the selection of target, strategy, approach, or model. In relation to efforts to improve the culture of e-government-based public services in the city of Magelang, several policy recommendations have been formulated, including: 1) preparation of a clear policy basis or special legal umbrella as a form of leadership commitment to the development of e-government in the city of Magelang so that it can be integrated into every OPD; 2) formulation of strategies related to the provision of Internet infrastructure and e-government applications for each OPD; 3) the Magelang City Government needs to allocate a number of resources, especially, financial, manpower, and time to build e-government in the framework of bureaucratic reform with a crosssectoral spirit.

6. CONCLUSION

The present study sought to investigate the information technology culture of public organizations in the context of public services in the Magelang City Government, Indonesia, and identify the driving and inhibiting factors of e-government based public services in the city. Findings suggest that Magelang's readiness level in information culture is strongly influenced by several types of culture, such as technocratic utopianism, anarchy, feudalism, dictatorship, and federalism. The most dominant information culture in the Magelang City Government is a dictatorship, marked by the standing of working leaders who always decide and control the existence of information in the organization, determine who may gain access, and control structure and related management reporting. Aside from physical issues, the main obstacle in developing a fine information culture in Magelang is restricted Internet extend and network, along with minimum hardware and human resources with ICT mastery.

This study has limitations, only looking at the information culture in the Magelang City Government, Indonesia. In the future it is necessary to conduct further research on the level of readiness or e-readiness of the e-government implementation model in the city of Magelang. To reduce the dominance of the dictatorial type of information culture model in the city of Magelang, it is necessary to consider the use of a hybrid model, in which the processes and activities of information technology management are centralized, while others use a scattered model (decentralized) (Canals et al., 2005; Indrajit, 2012; Wang et al., 2016). Furthermore, to overcome the limitations of human resources, the Magelang City Government can carry out internships to several universities and vocational schools around the city of Magelang on an ongoing basis that have information technology qualifications, besides that it is necessary to optimize cooperation with third parties in the e-government development process.

REFERENCES

- 1. Abbassy, M. M., & Mesbah, S. (2016). Effective e-Government and citizens adoption in Egypt. *International Journal of Computer Applications, 133*(7), 7–13. https://doi.org/10.5120/ijca2016907886
- Al-Masaeed, S., & Love, S. (2014). Implementing mobile government in Jordan: Strategies for success. *International Journal of Electronic Governance*, 7(1), 3–26. https://doi.org/10.1504/IJEG.2014.065077
 Al-Sarabi, A. H. A., & Khanfar, S. M. (2010). Electronic government services and benefits in the private and off.
- 3. Al-Sarabi, A. H. A., & Khanfar, S. M. (2010). Electronic government services and benefits in the private and public context: Saudi Arabia case study. *Interdisciplinary Journal of Contemporary Research in Business, 2*(7), 155–172. Retrieved from https://www.researchgate.net/publication/215826595_Electronic_Government _Services_and_Benefits_in_the_Private_and_Public_ContextSaudi_Arabia_Case_Study
- 4. Altman, Y., & Baruch, Y. (1998). Cultural theory and organizations: Analytical method and cases. *Organization Studies*, *19*(5), 769–785. https://doi.org/10.1177/017084069801900503
- 5. Amegavi, G. B., Bawole, J. N., & Buabeng, T. (2018). The dynamics of e-government enactment in a developing country public sector organisation: Evidence from Ghana. *International Journal of Electronic Governance, 10*(1), 74–92. https://doi.org/10.1504/IJEG.2018.091267
- 6. Badan Perencanaan Pembangunan Daerah (Bappeda). (2020). *Laporan Akhir Magelang Smart City* [Final report of Magelang Smart City]. Yogyakarta, Indonesia: Bappeda.
- 7. Balsamo, A. (1996). Myths of information: The cultural impact of new information technologies. *Technology Analysis & Strategic Management, 8*(3), 341–348. https://doi.org/10.1080/09537329608524256
- 8. Basu, S. (2004). E-government and developing countries: An overview. *International Review of Law, Computers and Technology, 18*(1), 109–132. https://doi.org/10.1080/13600860410001674779
- 9. Batara, E., Nurmandi, A., Warsito, T., & Pribadi, U. (2017). Are government employees adopting local e-government transformation? The need for having the right attitude, facilitating conditions and performance expectations. *Transforming Government: People, Process and Policy, 11*(4), 612–638. https://doi.org/10.1108/TG-09-2017-0056
- 10. Berryman, J. (2004). E-government: Issues and implications for public libraries. *Australian Library Journal*, 53(4), 349–359. https://doi.org/10.1080/00049670.2004.10721682
- 11. Borah, S. K. (2013). Implementation of citizen-centric e-Governance projects in Assam. *IOSR Journal of Humanities and Social Science*, *7*(2), 27–34. https://doi.org/10.9790/0837-0722734
- 12. Bowen, P., & Rose, R., & Pilkington, A. (2017). Mixed Methods-theory and practice sequential, explanatory approach. *International Journal of Quantitative and Qualitative Research Methods*, 5(2), 10–27. Retrieved from https://www.eajournals.org/wp-content/uploads/Mixed-Methods-Theory-and-Practice.-Sequential-Explanatory-Approach.pdf
- 13. Brewer, G. A., Neubauer, B. J., & Geiselhart, K. (2006). Designing and implementing e-government systems: Critical implications for public administration and democracy. *Journal of Administration & Society, 38*(4), 472–499. https://doi.org/10.1177/0095399706290638
- 14. Bulinska-Stangrecka, H. (2016). The impact of information and communication technologies on e-culture. *Global Management Journal, 8*, 14–20. Retrieved from https://www.researchgate.net/publication/318099424 _The_Impact_of_Information_and_Communication_Technologies_on_E-culture

VIRTUS

- 15. Bungin, B. (2020). Post-qualitative social research methods: Kuantitatif-kualitatif-mixed methods, positivism, postpositivism-phenomenology-postmodern, filsafat, paradigma, teori. Jakarta, Indonesia: Kencana.
- 16. Canals, A., Boisot, M., & MacMillan, I. (2005). Simulating I-Space (SimISpace): An agent-based approach to modeling knowledge flows (IN3 Working Paper Series WP04-006). Retrieved from https://www.uoc.edu /in3/dt/eng/wp04006.pdf
- 17. Chen, W., & Bennett, D. (2010). Managing wireless networks in the healthcare sector: Emerging experiences of
- cultural impacts. *Journal of E-Health Management*, *3*, 1–9. https://doi.org/10.5171/2010.500498 Côté-Leclerc, F., Boileau Duchesne, G., Bolduc, P., Gélinas-Lafrenière, A., Santerre, C., Desrosiers, J., & Levasseur, M. (2017). How does playing adapted sports affect quality of life of people with mobility limitations? 18. Côté-Leclerc, F., Results from a mixed-method sequential explanatory study. Health and Quality of Life Outcomes, 15(1), 1-8. https://doi.org/10.1186/s12955-017-0597-9
- Creswell, J. W., & Creswell, J. D. (2018). Research design qualitative, quantitative, and mixed methods approaches 19. (5th ed.). Los Angeles, CA: SAGE Publications.
- Creswell, J. W., & Piano Clark, V. L. (2018). Mendesain dan melaksanakan mixed methods research (2nd ed.). 20. Yogyakarta, Indonesia: Pustaka Pelajar.
- Da Veiga, A. (2016). Comparing the information security culture of employees who had read the information 21. security policy and those who had not: Illustrated through an empirical study. Information and Computer Security, 24(2), 139-151. https://doi.org/10.1108/ICS-12-2015-0048
- Elbahnasawy, N. G. (2014). E-government, Internet adoption and corruption: An empirical investigation. World 22. Development, 57, 114-126. https://doi.org/10.1016/j.worlddev.2013.12.005
- 23. Evans, D., & Yen, D. C. (2006). E-government: Evolving relationship of citizens and government, domestic and international development. Government Information Quarterly, 23(2), 207-235. https://doi.org/10.1016 /j.giq.2005.11.004
- Evanschitzky, H., & Iyer, G. R. (2007). E-services: Opportunities and threats. Deutscher Universitäts-Verlag. 24. https://doi.org/10.1007/978-3-8350-9614-1
- 25. Fang, S.-C., & Wang, J.-F. (2006). Effect of organizational culture and learning on manufacturing strategy selection: An empirical study. International Journal Management, 23(3), 503–514. Retrieved from https://cutt.ly/rPuA0jW
- 26. Fang, Z. (2002). E-government in digital era: Concept, practice and development. *International Journal of the Computer, the Internet and Management, 10*(2), 1–22. Retrieved from https://www.academia.edu/5231621 /E_Government_in_Digital_Era_Concept_Practice_and_Development
- 27. Ganapati, S., & Reddick, C. G. (2012). Open e-government in U.S. state governments: Survey evidence from chief information officers. Government Information Quarterly, 29(2), 115-122. https://doi.org/10.1016/j.giq.2011.09.006
- 28. Gilbert, D., Balestrini, P., & Littleboy, D. (2004). Barriers and benefits in the adoption of e-government. International Journal of Public Sector Management, 17(4), 286–301. https://doi.org/10.1108/09513550410539794
- 29. Ginman, M. (1987). Information culture and business performance. Iatul Quarterly, 2(2), 93-106. Retrieved from https://docs.lib.purdue.edu/cgi/viewcontent.cgi?article=1198&context=iatul
- 30. Gupta, B., Dasgupta, S., & Gupta, A. (2008). Adoption of ICT in a government organization in a developing The Journal of Strategic Information Systems, An empirical study. country: 17(2),140-154. https://doi.org/10.1016/j.jsis.2007.12.004
- Gupta, K. P., Bhaskar, P., & Singh, S. (2016). Critical factors influencing e-government adoption in India: An investigation of the citizens' perspectives. *Journal of Information Technology Research*, 9(4), 28-44. https://doi.org/10.4018/JITR.2016100103 31.
- 32. Gupta, K. P., Singh, S., & Bhaskar, P. (2016). Citizen adoption of e-government: A literature review and conceptual framework. Electronic Government, an International Journal, 12(2), 160-185. https://doi.org '10.1504/EG.2016.076134
- 33. Gupta, K. P., Singh, S., & Bhaskar, P. (2018). Citizens' perceptions on benefits of e-governance services. International Journal of Electronic Governance, 10(1), 24-55. https://doi.org/10.1504/IJEG.2018.091261
- 34. Hansen, P., & Widén, G. (2016). The embeddedness of collaborative information seeking in information culture. Journal of Information Science, 43(4), 554–566. https://doi.org/10.1177/0165551516651544
- 35. Harper, G. R. & Utley, D. R. (2001). Organizational culture and successful information technology implementation. Engineering Management Journal, 13(2), 11-15. https://doi.org/10.1080/10429247.2001.11415111
- 36. Ho, A. T.-K. (2002). Reinventing local government and the e-government initiative. Public Administration Review, 62(4), 434-444. https://doi.org/10.1111/0033-3352.00197
- 37. Hofstede, G. (1981). Culture and organizations. International Studies of Management & Organization, 10(4), 15-41. https://doi.org/10.1080/00208825.1980.11656300
- 38. Hooda, A., & Singla, M. L. (2021). Core-competencies — a key to future-oriented and sustainable e-governance implementation: A mixed method research. Transforming Government: People, Process and Policy, 15(1), 80-107. https://doi.org/10.1108/TG-12-2019-0122
- 39. Hughes, O. E. (1994). Public management and administration: An introduction. New York, NY: Palgrave MacMillan.
- 40. Hung, S.-Y., Chang, C.-M., & Kuo, S.-R. (2013). User acceptance of mobile e-government services: An empirical study. Government Information Quarterly, 30(1), 33-44. https://doi.org/10.1016/j.giq.2012.07.008
- Indonesian Internet Service Providers Association (APJII). (2017). Penetrasi & perilaku pengguna internet Indonesia. 41. Retrieved from https://apjii.or.id/content/read/39/342/Hasil-Survei-Penetrasi-dan-Perilaku-Pengguna-Internet-Indonesia-2017
- 42. Indrajit, R. E. (2002). Electronic government, strategi pembangunan dan pengembangan sistem pelayanan publik berbasis teknologi digital. Yogyakarta, Indonesia: Andi Offset.
- 43. Indrajit, R. E. (2005). E-government in action ragam. Kasus implementasi sukses di berbagai belahan dunia. Yogyakarta, Indonesia: Andi Offset.
- Indrajit, R. E. (2012). Taksonomi budaya informasi dalamkonteks organisasi dan korporasi. BUDAYA TIK, 006. 44. Retrieved from https://www.academia.edu/11625223/Ekoji999_Edisi006_14Sep12_Budaya_Organisasi
- Ivankova, N. V., Creswell, J. W., & Stick, S. L. (2006). Using mixed-methods sequential explanatory design: From 45. theory to practice. *Field Methods*, *18*(1), 3–20. https://doi.org/10.1177/1525822X05282260 Javidian, Z., & Mollayaaghobi, S. S. (2011). Evaluating & improving electronic trust factors in e-government (Case
- 46. study: Mashhad electronic city). Interdisciplinary Journal of Contemporary Research in Business, 3(4), 126-135. Retrieved from https://cutt.ly/SPuLhdk

VIRTUS

- 47. Jenkins, D. M. (2019). What the best leadership educators do: A sequential explanatory mixed methods study of instructional and assessment strategy use in leadership education. *Journal of Leadership Education*, *19*(4), 37–55. https://doi.org/10.12806/V19/I4/R4
- Johnson, M. (2012). A study of e-market adoption barriers in the local government sector. *Journal of Enterprise Information Management*, 25(6), 509–536. https://doi.org/10.1108/17410391211272810
- Kalsi, N. S., & Kiran, R. (2013). E-governance success factors: An analysis of e-governance initiatives of ten major states of India. *International Journal of Public Sector Management*, 26(4), 320–336. https://doi.org/10.1108/IJPSM-08-2011-0101
- 50. Kalsi, N. S., Kiran, R., & Vaidya, S. C. (2008). Effective e-governance for good governance in India. *International Journal of Business Research Papers*, 5(6), 212–229. Retrieved from https://www.researchgate.net/publication /228949741_Effective_e-Governance_for_Good_Governance_in_India
- 51. Kalsi, N. S., Kiran, R., & Vaidya, S. C. (2009). Changing perspectives and pragmatics of good governance and e-governance in India: A share vision of citizens. *International Journal of E-Governance, 2*(2), 251–271. https://doi.org/10.1504/IJEG.2009.029133
- 52. Kass, T. (1991). Culture and technology. *Technology Analysis & Strategic Management, 3*(3), 285-288. https://doi.org/10.1080/09537329108524057
- 53. Katz, J. P., & Townsend, J. B. (2000). The role of information technology in the "fit" between culture, business strategy and organizational structure of global firms. *Journal of Global Information Management, 8*(2). https://doi.org/10.4018/jgim.2000040102
- 54. Khamis, M. M., & van der Weide, T. P. (2017). Conceptual diagram development for sustainable e-government implementation. *The Electronic Journal of E-Government, 15*(1), 33–43. Retrieved from https://repository.ubn.ru.nl/bitstream/handle/2066/173177/173177.pdf?sequence=1&isAllowed=y
- 55. Krohwinkel, A. (2015). A configurational approach to project delays: Evidence from a sequential mixed methods study. *Journal of Mixed Methods Research*, *9*(4), 335–361. https://doi.org/10.1177/1558689814522709
- 56. Lau, T. Y., Aboulhoson, M., Lin, C., & Atkin, D. J. (2008). Adoption of e-government in three Latin American countries: Argentina, Brazil and Mexico. *Telecommunications Policy*, *32*(2), 88–100. https://doi.org/10.1016/j.telpol.2007.07.007
- 57. Leech, N. L., & Onwuegbuzie, A. J. (2007). An array of qualitative data analysis tools: A call for data analysis triangulation. *School Psychology Quarterly*, *22*(4), 557–584. https://doi.org/10.1037/1045-3830.22.4.557
- 58. Leech, N. L., & Onwuegbuzie, A. J. (2009). A typology of mixed methods research designs. *Quality and Quantity*, *43*(2), 265–275. https://doi.org/10.1007/s11135-007-9105-3
- 59. Lubbad, A. R. M., & Bohari, A. M. B. (2016). The conceptual perspective of the moderating role of organizational culture in the information technology innovativeness and adoption decision. *Information and Knowledge Management*, 6(7), 80–87. Retrieved from https://www.iiste.org/Journals/index.php/IKM/article /view/31732/32604
- 60. Luna-Reyes, L. F., & Gil-Garcia, J. R. (2011). Using institutional theory and dynamic simulation to understand complex e-Government phenomena. *Government Information Quarterly, 28*(3), 329–345. https://doi.org /10.1016/j.giq.2010.08.007
- 61. Maleku, A., Kim, Y. K., Kagotho, N., & Lim, Y. (2021). Expanding the transformative explanatory sequential mixed methods design archetype in a cross-cultural context: The polemics of African refugee livelihoods in places of resettlement. *Journal of Mixed Methods Research*, *15*(2), 212–239. https://doi.org/10.1177/1558689820936378
- 62. Mardiana, S., Tjakraatmadja, J. H., & Aprianingsih, A. (2018). How organizational culture affects information system success: The case of an Indonesia IT-based company. *Journal of Information Systems Engineering and Business Intelligence*, 4(2), 84–95. https://doi.org/10.20473/jisebi.4.2.84-95
- 63. McKim, C. A. (2017). The value of mixed methods research: A mixed methods study. *Journal of Mixed Methods Research*, *11*(2), 202-222. https://doi.org/10.1177/1558689815607096
- 64. Miles, M. B., & Huberman, A. M. (1994). *Qualitative data analysis: An expanded sourcebook* (2nd ed.). SAGE Publications. Retrieved from https://vivauniversity.files.wordpress.com/2013/11/milesandhuberman1994.pdf
- 65. Moon, M. J. (2002). The evolution of e-government among municipalities: Rhetoric or reality? *Public Administration Review*, *62*(4), 424-433. https://doi.org/10.1111/0033-3352.00196
- 66. Morren, M., Gelissen, J. P. T. M., & Vermunt, J. K. (2013). Exploring the response process of culturally differing survey respondents with a response style: A sequential mixed methods study. *Field Methods*, 25(2), 162–181. https://doi.org/10.1177/1525822X12453535
- 67. Muhaimin, M., Habibi, A., Mukminin, A., Saudagar, F., Pratama, R., Wahyuni, S., ... Indrayana, B. (2019). A sequential explanatory investigation of TPACK: Indonesian science teachers' survey and perspective. *Journal of Technology and Science Education*, *9*(3), 269–281. http://doi.org/10.3926/jotse.662
- 68. Nugraha, J. T. (2018). E-government dan Pelayanan Publik (Studi Tentang Elemen Sukses Pengembangan e-government di Pemerintah Kabuaten Sleman). *Jurnal Komunikasi dan Kajian Media, 2*(1). Retrieved from https://jurnal.untidar.ac.id/index.php/komunikasi/article/view/758
- 69. Olasina, G. (2014). E-parliament services as tools for anti-corruption and transparency. *International Journal of Electronic Governance*, *7*(1), 27–55. https://doi.org/10.1504/IJEG.2014.065083
- 70. Oliver, G. (2017). Understanding information culture: Conceptual and implementation issues. *Journal of Information Science Theory and Practice*, *5*(1), 6–14. https://doi.org/10.1633/JISTaP.2017.5.1.1
- 71. Oliver, G., & Foscarini, F. (2014). *Records management and information culture: Tackling the people problem*. London. The UK: Facet Publishing.
- 72. Oni, A., Okunoye, A., & Mbarika, V. (2016). Evaluation of e-government implementation: The case of state government websites in Nigeria. *The Electronic Journal of E-Government, 14*(1), 48–59. Retrieved from https://www.proquest.com/docview/1804470541?pq-origsite=primo
- 73. Pederson, K. (2016). eGovernment in local government: Challenges and capabilities. *The Electronic Journal of e-Government*, 14(1), 99–116. Retrieved from https://vbn.aau.dk/ws/portalfiles/portal/249045891 /ejeg_volume14_issue1_article448.pdf
- 74. Raharjo, B. (2001). Membangun e-government. Paper presented at the National Seminar on Computer Networking II.
- Rana, N. P., Dwivedi, Y. K., Lal, B., Williams, M. D., & Clement, M. (2017). Citizens' adoption of an electronic government system: Towards a unified view. *Information Systems Frontiers*, 19(3), 549–568. https://doi.org /10.1007/s10796-015-9613-y

VIRTUS

- 76. Rashid, M. A. A., Othman, M. N. A., & Narehan, H. (2011). Factors determining user adoption of e-Government at the state level in Malaysia: A case study. *International Journal of Arts & Sciences*, 4(26), 205–213. Retrieved from https://www.proquest.com/openview/03b2f6fe05ad85edc0cb967145d8ea9d/1?pq-origsite=gscholar&cbl=626342
- 77. Rehman, M., Esichaikul, V., & Kamal, M. (2011). Factors influencing e-government adoption in Pakistan. *Transforming Government: People, Process and Policy, 6*(3), 258–282. https://doi.org/10.1108/17506161211251263
- 78. Saxena, S. (2005). E-governance and good governance: The Indian context. *The Indian Journal of Political Science*, *66*(2), 313–328. Retrieved from https://www.jstor.org/stable/41856130
- 79. Schein, E. H. (1992). Culture and related corporate realities. Homewood, CA: Richard D. Irwin Inc.
- 80. Shachaf, P. (2008). Cultural diversity and information and communication technology impacts on global virtual teams: An exploratory study. *Information & Management*, *45*(2), 131–142. https://doi.org/10.1016/j.im.2007.12.003
- Sharma, A. (1994). Organizational culture and adoption of high-technology products. *Journal of Marketing Management*, *10*(6), 513-526. https://doi.org/10.1080/0267257X.1994.9964297
 Sharma, P. (2016). Facts and abstracts of e-governance in India. *International Journal of Commerce and Management*
- *Research, 2*(11), 34–37. Retrieved from http://www.managejournal.com/archives/2016/vol2/issue11/2-9-63
- Sheth, J. N., & Sharma, A. (2007). E-Services A framework for growth. In H. Evanschitzky & G. R. Iyer (Eds.), *E-services: Opportunities and threats.* (pp. 7-12). Deutscher Universitats-Verlag. https://doi.org/10.1007/978-3-8350-9614-1_2
- 84. Sijabat, R. (2020). Analysis of e-government services: A Study of the adoption of electronic tax filing in Indonesia. *Jurnal Ilmu Sosial Dan Ilmu Politik, 23*(3), 179. https://doi.org/10.22146/jsp.52770
- 85. Slamet, Hamdan, A. R. B., & Deraman, A. (2007). Restrukturisasi dan reorientasi menuju percepatan target e-government di Indonesia. Paper presented at *Konferensi Nasional Sistem Informasi 2007*. Retrieved from http://repository.uin-malang.ac.id/435/1/01-Restrukturisasi.pdf
- Srivastava, C. S., & Teo, T. S. H. (2010). E-government, e-business and national economic performance. *Communications of the Association for Information Systems*, *26*(14), 267–286. https://doi.org/10.17705/1CAIS.02614
 Stake, R. E. (1995). *The art of case study research*. Thousand Oaks, CA: SAGE Publications.
- State, R. E. (195). The latter of case state research information of a by business process change A methodology for public sector. *International Journal of Information Management*, 27(4), 221–232. https://doi.org/10.1016 /j.ijinfomgt.2007.02.006
- 89. Thompson, M., & Wildavsky, A. (1986). A cultural theory of information bias in organizations. *Journal of Management Studies*, *23*(3), 273–286. https://doi.org/10.1111/j.1467-6486.1986.tb00954.x
- 90. Tse, T., Tiong, J., & Kangaslahti, V. (2004). The effect of cultural norms on the uptake of information and communication technologies in Europe: A conceptual analysis. *International Journal of Management, 21*(3), 382–392. Retrieved from https://www.proquest.com/docview/233230593?pq-origsite=gscholar&fromopenview=true
- 91. Turner, J. H. (1998). The structure of sociological theory (6th ed.). Belmont, CA: Wadsworth Publishing Company.
- 92. Upward, F., Reed, B., Oliver, G., & Evans, J. (2013). Recordkeeping informatics: Re-figuring a discipline in crisis with a single minded approach. *Records Management Journal*, 23(1), 37–50. https://doi.org/10.1108 /09565691311325013
- 93. Urbina, A. U., & Abe, N. (2017). Citizen-centric Perspective on the adoption of e-government in the Philippines. *The Electronic Journal of E-Government, 15*(2), 63–83. Retrieved from https://academic-publishing.org /index.php/ejeg/article/view/641
- 94. Wang, W.-T., Wang, Y.-S., & Liu, E.-R. (2016). The stickiness intention of group-buying websites: The integration of the commitment-trust theory and e-commerce success model. *Information and Management*, 53(5), 625–642. https://doi.org/10.1016/j.im.2016.01.006
- 95. Weerakkody, V., Janssen, M., & Dwivedi, Y. K. (2011). Transformational change and business process reengineering (BPR): Lessons from the British and Dutch public sector. *Government Information Quarterly*, *28*(3), 320-328. https://doi.org/10.1016/j.giq.2010.07.010
- 96. Widén, G., & Hansen, P. (2012). Managing collaborative information sharing: Bridging research on information culture and collaborative information behaviour. *Information Research*, *17*(4). Retrieved from http://www.informationr.net/ir/17-4/paper538.html#.YgpMXN9BxPZ
- 97. Widén-Wulff, G. (2000). Business information culture: A qualitative study of the information culture in the Finnish insurance industry. *Information Research*, 5(3), 3–5. Retrieved from http://informationr.net/ir/5-3/paper77.html
- 98. Winarsih., R. A. S. (2008). Manajemen pelayanan, pengembangan model konseptual, penerapan citizen's charter dan standar pelayanan minimal. Yogyaka, Indonesia: Pustaka Pelajar.
- 99. Xu, H. (2012). Information technology, public administration and citizen participation: The impacts of e-government on political and administrative processes. *Public Administration Review*, *72*(6), 915–920. https://doi.org/10.1111/j.1540-6210.2012.02671.x
- 100. Zahid, H., & Din, B. H. (2019). Determinants of intention to adopt e-government services in Pakistan: An imperative for sustainable development. *Resources, 8*(128), 1–25. https://doi.org/10.3390/resources8030128

VIRTUS

APPENDIX A. SPBE ASSESSMENT BY THE UNITED NATIONS 2012–2018 FOR INDONESIA

| No. | Description | 2012 | 2014 | 2016 | 2018 |
|-----|--------------------------------------|--------|--------|--------|--------|
| 1. | Rank | 97 | 106 | 116 | 107 |
| 2. | SPBE Development Index | 0.4949 | 0.4487 | 0.4478 | 0.5258 |
| 3. | Online Services Index | 0.4967 | 0.3622 | 0.3623 | 0.5694 |
| 4. | Telecommunication Connectivity Index | 0.1897 | 0.3054 | 0.3016 | 0.3222 |
| 5. | Human Capital Index | 0.7982 | 0.6786 | 0.6796 | 0.6857 |

APPENDIX B. STAGE OF IMPLEMENTING EXPLANATORY SEQUENTIAL DESIGN

| Step | Phase | Procedure | Results |
|--|--|--|--|
| Step 1Designing and implementing a quantitative strand: a. Quantitative research questions and selection of the quantitative approach; b. Obtaining permits; c. Identify quantitative samples; d. Collecting closed-ended data with the instrument; e. Analyze quantitative data using descriptive statistics, to answer quantitative research questions and facilitate participants for the second phase. | a. Quantitative data collection; b. Quantitative data analysis; c. Case selection: interview protocol development. | a. Survey (N=30); b. Screening data; c. Frequency snalysis; d. SPSS software; e. Selecting participants from each regional apparatus organization (OPD) based on unique answers and the principle of maximum variation; f. Develop interview questions. | a. Numerical data; b. Descriptive statistics; c. Factor loading; d. Presentation of data (tables, frequency distribution tables, graphs, diagrams); e. Central symptom measurement (mode, median, mean). |
| Step 2 Using strategies to move from quantitative results: a. Determine the results to be explained, for example, significant results, non- significant results, outliers, group differences; b. Use quantitative results to refine qualitative and mixed methods questions, determine which participants to select for the qualitative sample, and design qualitative data collection protocols. | Qualitative data analysis | a. Interview with participants; b. Interview materials; c. Documents. | a. Text data, interview transcripts, documents; b. Image data, photos. |
| Step 3Designing and implementingqualitative strands:a. Stating the researchquestions following thequantitative results and definingthe qualitative approach;b. Obtaining permits;c. Selecting a qualitativesample that can explain thequantitative results;d. Collecting open-ended datawith quantitative results-guidedprotocols;e. Analyzing qualitative datausing theme developmentprocedures and specificprocedures for qualitativeapproaches to answerqualitative questions anddetermine mixed methods. | Qualitative data collection | a. Theme analysis; b. In-case and cross-case theme development; c. Cross-theme analysis. | a. Multiple case analysis; b. Same and different themes and categories. |
| Step 4Interpreting connected results:a. Summarizingandinterpreting quantitative results;b. Summarizingandinterpreting qualitative results;c. Discussingthe extentandin what ways qualitative results.help explain quantitative results. | Combination of quantitative and qualitative results | a. Interpretation and explanation of quantitative and qualitative results. | a. Discussion; b. Implications; c. Further research. |

APPENDIX C. TECHNOCRATIC UTOPIANISM

Table C.1. Digital technology-based services will become more effective and efficient

| No. | Respondent's perception | Frequency | Percentage | Valid percentage | Cumulative percentage |
|-----|-------------------------|-----------|------------|------------------|-----------------------|
| 1 | Agreement | 15 | 50.0 | 50.0 | 50.0 |
| 2 | Strongly agreement | 14 | 46.6 | 46.6 | 96.6 |
| 3 | No answer | 1 | 3.33 | 0 | 100.0 |
| 4 | Disagreement | 0 | 0 | 0 | |
| 5 | Strong disagreement | 0 | 0 | 0 | |
| | Total | 30 | 100.0 | 100.0 | |

Table C.2. Magelang City Government makes rules regarding digital-based services

| No. | Respondent's perception | Frequency | Percentage | Valid percentage | Cumulative percentage |
|-----|-------------------------|-----------|------------|------------------|-----------------------|
| 1 | Agreement | 20 | 66.6 | 66.6 | 66.6 |
| 2 | Strongly agreement | 8 | 26.6 | 26.6 | 96.6 |
| 3 | No answer | 2 | 6.6 | 6.6 | 100.0 |
| 4 | Disagreement | 0 | 0 | 0 | |
| 5 | Strong disagreement | 0 | 0 | 0 | |
| | Total | 30 | 100.0 | 100.0 | |

Table C.3. Magelang City Government held training for website managers

| No. | Respondent's perception | Frequency | Percentage | Valid percentage | Cumulative percentage |
|-----|-------------------------|-----------|------------|------------------|-----------------------|
| 1 | Agreement | 15 | 50.0 | 50.0 | 50.0 |
| 2 | Strongly agreement | 12 | 40.0 | 40.0 | 90.0 |
| 3 | No answer | 3 | 10.0 | 10.0 | 100.0 |
| 4 | Disagreement | 0 | 0 | 0 | |
| 5 | Strong disagreement | 0 | 0 | 0 | |
| | Total | 30 | 100.0 | 100.0 | |

Table C.4. OPD website managers have ICT competence

| No. | Respondent's perception | Frequency | Percentage | Valid percentage | Cumulative percentage |
|-----|-------------------------|-----------|------------|------------------|-----------------------|
| 1 | Agreement | 16 | 53.3 | 53.3 | 53.3 |
| 2 | Strongly agreement | 12 | 40.0 | 40.0 | 96.6 |
| 3 | No answer | 2 | 6.6 | 6.6 | 100.0 |
| 4 | Disagreement | 0 | 0 | 0 | |
| 5 | Strong disagreement | 0 | 0 | 0 | |
| | Total | 30 | 100.0 | 100.0 | |

Table C.5. Allocation of funds for e-government development

| No. | Respondent's perception | Frequency | Percentage | Valid percentage | Cumulative percentage |
|-----|-------------------------|-----------|------------|------------------|-----------------------|
| 1 | Agreement | 18 | 60.0 | 60.0 | 60.0 |
| 2 | Strongly agreement | 10 | 33.3 | 33.3 | 93.3 |
| 3 | No answer | 2 | 6.6 | 6.6 | 100.0 |
| 4 | Disagreement | 0 | 0 | 0 | |
| 5 | Strong disagreement | 0 | 0 | 0 | |
| | Total | 30 | 100.0 | 100.0 | |

Table C.6. Cooperation with third parties in e-government development

| No. | Respondent's perception | Frequency | Percentage | Valid percentage | Cumulative percentage |
|-----|-------------------------|-----------|------------|------------------|-----------------------|
| 1 | Agreement | 22 | 73.3 | 73.3 | 73.3 |
| 2 | Strongly agreement | 5 | 16.6 | 16.6 | 89.9 |
| 3 | No answer | 3 | 10.0 | 10.0 | 100.0 |
| 4 | Disagreement | 0 | 0 | 0 | |
| 5 | Strong disagreement | 0 | 0 | 0 | |
| | Total | 30 | 100.0 | 100.0 | |

Table C.7. Building infrastructure for e-government development

| No. | Respondent's perception | Frequency | Percentage | Valid percentage | Cumulative percentage |
|-----|-------------------------|-----------|------------|------------------|-----------------------|
| 1 | Agreement | 20 | 66.6 | 66.6 | 66.6 |
| 2 | Strongly agreement | 9 | 30.0 | 30.0 | 96.6 |
| 3 | No answer | 1 | 3.3 | 3.3 | 100.0 |
| 4 | Disagreement | 0 | 0 | 0 | |
| 5 | Strong disagreement | 0 | 0 | 0 | |
| | Total | 30 | 100.0 | 100.0 | |

Table C.8. Respondent's nescience on the benefits of e-government in public services

| No. | Respondent's perception | Frequency | Percentage | Valid percentage | Cumulative percentage |
|-----|-------------------------|-----------|------------|------------------|-----------------------|
| 1 | Agreement | 0 | 0 | 0 | 0 |
| 2 | Strongly agreement | 0 | 0 | 0 | 0 |
| 3 | No answer | 2 | 6.6 | 6.6 | 6.6 |
| 4 | Disagreement | 20 | 66.6 | 66.6 | 73.2 |
| 5 | Strong disagreement | 8 | 26.6 | 26.6 | 100.0 |
| | Total | 30 | 100.0 | 100.0 | |

VIRTUS 76

APPENDIX D. ANARCHY

Table D.1. Leaders make special policies on information management

| No. | Respondent's perception | Frequency | Percentage | Valid percentage | Cumulative percentage |
|-----|-------------------------|-----------|------------|------------------|-----------------------|
| 1 | Agreement | 21 | 70.0 | 70.0 | 70.0 |
| 2 | Strongly agreement | 6 | 20.0 | 20.0 | 90.0 |
| 3 | No answer | 3 | 10.0 | 10.0 | 100.0 |
| 4 | Disagreement | 0 | 0 | 0 | |
| 5 | Strong disagreement | 0 | 0 | 0 | |
| | Total | 30 | 100.0 | 100.0 | |

Table D.2. Standard operation procedure on information management in each OPD

| No. | Respondent's perception | Frequency | Percentage | Valid percentage | Cumulative percentage |
|-----|-------------------------|-----------|------------|------------------|-----------------------|
| 1 | Agreement | 22 | 73.3 | 73.3 | 73.3 |
| 2 | Strongly agreement | 6 | 20.0 | 20.0 | 93.3 |
| 3 | No answer | 2 | 6.6 | 6.6 | 100.0 |
| 4 | Disagreement | 0 | 0 | 0 | |
| 5 | Strong disagreement | 0 | 0 | 0 | |
| | Total | 30 | 100.0 | 100.0 | |

Table D.3. Access to public information through website

| No. | Respondent's perception | Frequency | Percentage | Valid percentage | Cumulative percentage |
|-----|-------------------------|-----------|------------|------------------|-----------------------|
| 1 | Agreement | 18 | 60.0 | 60.0 | 60.0 |
| 2 | Strongly agreement | 11 | 36.6 | 36.6 | 96.6 |
| 3 | No answer | 1 | 3.3 | 3.3 | 100.0 |
| 4 | Disagreement | 0 | 0 | 0 | |
| 5 | Strong disagreement | 0 | 0 | 0 | |
| | Total | 30 | 100.0 | 100.0 | |

Table D.4. Involving outourcing personnel in website management

| No. | Respondent's perception | Frequency | Percentage | Valid percentage | Cumulative percentage |
|-----|-------------------------|-----------|------------|------------------|-----------------------|
| 1 | Agreement | 14 | 46.6 | 46.6 | 46.6 |
| 2 | Strongly agreement | 3 | 10.0 | 10.0 | 56.6 |
| 3 | No answer | 6 | 20.0 | 20.0 | 76.6 |
| 4 | Disagreement | 7 | 23.3 | 23.3 | 100.0 |
| 5 | Strong disagreement | 0 | 0 | 0 | |
| | Total | 30 | 100.0 | 100.0 | |

Table D.5. Not all important information can be accessed by the public

| No. | Respondent's perception | Frequency | Percentage | Valid percentage | Cumulative percentage |
|-----|-------------------------|-----------|------------|------------------|-----------------------|
| 1 | Agreement | 18 | 60.0 | 60.0 | 60.0 |
| 2 | Strongly agreement | 4 | 13.3 | 13.3 | 73.3 |
| 3 | No answer | 3 | 10.0 | 10.0 | 83.3 |
| 4 | Disagreement | 4 | 13.3 | 13.3 | 96.6 |
| 5 | Strong disagreement | 1 | 3.3 | 3.3 | 100.0 |
| | Total | 30 | 100.0 | 100.0 | |

VIRTUS 77

APPENDIX E. FEUDALISM

Table E.1. The application of digital technology in public services is very important

| No. | Respondent's perception | Frequency | Percentage | Valid percentage | Cumulative percentage |
|-----|-------------------------|-----------|------------|------------------|-----------------------|
| 1 | Agreement | 16 | 53.3 | 53.3 | 53.3 |
| 2 | Strongly agreement | 13 | 43.3 | 43.3 | 96.6 |
| 3 | No answer | 1 | 3.3 | 3.3 | 100.0 |
| 4 | Disagreement | 0 | 0 | 0 | |
| 5 | Strong disagreement | 0 | 0 | 0 | |
| | Total | 30 | 100.0 | 100.0 | |

Table E.2. Web managers in every OPD must know information technology

| No. | Respondent's perception | Frequency | Percentage | Valid percentage | Cumulative percentage |
|-----|-------------------------|-----------|------------|------------------|-----------------------|
| 1 | Agreement | 15 | 50.0 | 50.0 | 50.0 |
| 2 | Strongly agreement | 14 | 46.6 | 46.6 | 96.6 |
| 3 | No answer | 1 | 3.3 | 3.3 | 100.0 |
| 4 | Disagreement | 0 | 0 | 0 | |
| 5 | Strong disagreement | 0 | 0 | 0 | |
| | Total | 30 | 100.0 | 100.0 | |

Table E.3. Digital technology facilitates task completion

| No. | Respondent's perception | Frequency | Percentage | Valid percentage | Cumulative percentage |
|-----|-------------------------|-----------|------------|------------------|-----------------------|
| 1 | Agreement | 16 | 53.3 | 53.3 | 53.3 |
| 2 | Strongly agreement | 12 | 40.0 | 40.0 | 93.3 |
| 3 | No answer | 2 | 6.6 | 6.6 | 100.0 |
| 4 | Disagreement | 0 | 0 | 0 | |
| 5 | Strong disagreement | 0 | 0 | 0 | |
| | Total | 30 | 100.0 | 100.0 | |

Table E.4. Web manager in every OPD with information technology education

| No. | Respondent's perception | Frequency | Percentage | Valid percentage | Cumulative percentage |
|-----|-------------------------|-----------|------------|------------------|-----------------------|
| 1 | Agreement | 16 | 53.3 | 53.3 | 53.3 |
| 2 | Strongly agreement | 8 | 26.6 | 26.6 | 79.9 |
| 3 | No answer | 2 | 6.6 | 6.6 | 86.5 |
| 4 | Disagreement | 4 | 13.3 | 13.3 | 100 |
| 5 | Strong disagreement | 0 | 0 | 0 | |
| | Total | 30 | 100.0 | 100.0 | |

Table E.5. All Magelang City Government employees are able to operate computers

| No. | Respondent's perception | Frequency | Percentage | Valid percentage | Cumulative percentage |
|-----|-------------------------|-----------|------------|------------------|-----------------------|
| 1 | Agreement | 13 | 43.3 | 43.3 | 43.3 |
| 2 | Strongly agreement | 5 | 16.6 | 16.6 | 59.9 |
| 3 | No answer | 6 | 20.0 | 20.0 | 79.9 |
| 4 | Disagreement | 6 | 20.0 | 20.0 | 100.0 |
| 5 | Strong disagreement | 0 | 0 | 0 | |
| | Total | 30 | 100.0 | 100.0 | |

Table E.6. Choosing to do tasks manually instead of using a computer

| No. | Respondent's perception | Frequency | Percentage | Valid percentage | Cumulative percentage |
|-----|-------------------------|-----------|------------|------------------|-----------------------|
| 1 | Agreement | 17 | 56.6 | 56.6 | 56.6 |
| 2 | Strongly agreement | 7 | 23.3 | 23.3 | 79.9 |
| 3 | No answer | 4 | 13.3 | 13.3 | 93.2 |
| 4 | Disagreement | 1 | 3.3 | 3.3 | 96.5 |
| 5 | Strong disagreement | 1 | 3.3 | 3.3 | 100.0 |
| | Total | 30 | 100.0 | 100.0 | |

VIRTUS 78

APPENDIX F. DICTATORSHIP

Table F.1. All information in the OPD must go through the leadership first

| No. | Respondent's perception | Frequency | Percentage | Valid percentage | Cumulative percentage |
|-----|-------------------------|-----------|------------|------------------|-----------------------|
| 1 | Agreement | 17 | 56.6 | 56.6 | 56.6 |
| 2 | Strongly agreement | 7 | 23.3 | 23.3 | 79.9 |
| 3 | No answer | 3 | 10.0 | 10.0 | 89.9 |
| 4 | Disagreement | 3 | 10.0 | 10.0 | 100.0 |
| 5 | Strong disagreement | 0 | 0 | 0 | |
| | Total | 30 | 100.0 | 100.0 | |

Table F.2. OPD leaders decide which information can be accessed by the public

| No. | Respondent's perception | Frequency | Percentage | Valid percentage | Cumulative percentage |
|-----|-------------------------|-----------|------------|------------------|-----------------------|
| 1 | Agreement | 18 | 60.0 | 60.0 | 60.0 |
| 2 | Strongly agreement | 9 | 30.0 | 30.0 | 90.0 |
| 3 | No answer | 2 | 6.6 | 6.6 | 96.6 |
| 4 | Disagreement | 1 | 3.3 | 3.3 | 100.0 |
| 5 | Strong disagreement | 0 | 0 | 0 | |
| | Total | 30 | 100.0 | 100.0 | |

Table F.3. There is a special information management unit in each OPD

| No. | Respondent's perception | Frequency | Percentage | Valid percentage | Cumulative percentage |
|-----|-------------------------|-----------|------------|------------------|-----------------------|
| 1 | Agreement | 19 | 63.3 | 63.3 | 63.3 |
| 2 | Strongly agreement | 6 | 20.0 | 20.0 | 83.3 |
| 3 | No answer | 3 | 10.0 | 10.0 | 93.3 |
| 4 | Disagreement | 2 | 6.6 | 6.6 | 100.0 |
| 5 | Strong disagreement | 0 | 0 | 0 | |
| | Total | 30 | 100.0 | 100.0 | |

Table F.4. Leaders create information distribution system flow

| No. | Respondent's perception | Frequency | Percentage | Valid percentage | Cumulative percentage |
|-----|-------------------------|-----------|------------|------------------|-----------------------|
| 1 | Agreement | 22 | 73.3 | 73.3 | 73.3 |
| 2 | Strongly agreement | 6 | 20.0 | 20.0 | 93.3 |
| 3 | No answer | 2 | 6.6 | 6.6 | 100.0 |
| 4 | Disagreement | 0 | 0 | 0 | |
| 5 | Strong disagreement | 0 | 0 | 0 | |
| | Total | 30 | 100.0 | 100.0 | |

APPENDIX G. FEDERALISM

Table G.1. Leaders open information access flows to all employees

| No. | Respondent's perception | Frequency | Percentage | Valid percentage | Cumulative percentage |
|-----|-------------------------|-----------|------------|------------------|-----------------------|
| 1 | Agreement | 19 | 63.3 | 63.3 | 63.3 |
| 2 | Strongly agreement | 6 | 20.0 | 20.0 | 86.2 |
| 3 | No answer | 2 | 6.6 | 6.6 | 89.6 |
| 4 | Disagreement | 3 | 10.0 | 10.0 | 100.0 |
| 5 | Strong disagreement | 0 | 0 | 0 | |
| | Total | 30 | 100.0 | 100.0 | |

Table G.2. Leaders always share information and discuss with all employees

| No. | Respondent's perception | Frequency | Percentage | Valid percentage | Cumulative percentage |
|-----|-------------------------|-----------|------------|------------------|-----------------------|
| 1 | Agreement | 21 | 70.0 | 70.0 | 70.0 |
| 2 | Strongly agreement | 6 | 20.0 | 20.0 | 90.0 |
| 3 | No answer | 3 | 10.0 | 10.0 | 100.0 |
| 4 | Disagreement | 0 | 0 | 0 | |
| 5 | Strong disagreement | 0 | 0 | 0 | |
| | Total | 30 | 100.0 | 100.0 | |

Table G.3. Leaders debate regarding information governance with employees

| No. | Respondent's perception | Frequency | Percentage | Valid percentage | Cumulative percentage |
|-----|-------------------------|-----------|------------|------------------|-----------------------|
| 1 | Agreement | 24 | 80.0 | 80.0 | 80.0 |
| 2 | Strongly agreement | 6 | 20.0 | 20.0 | 100.0 |
| 3 | No answer | 0 | 0 | 0 | |
| 4 | Disagreement | 0 | 0 | 0 | |
| 5 | Strong disagreement | 0 | 0 | 0 | |
| | Total | 30 | 100.0 | 100.0 | |

Table G.4. Creating a special unit for information management in each OPD

| No. | Respondent's perception | Frequency | Percentage | Valid percentage | Cumulative percentage |
|-----|-------------------------|-----------|------------|------------------|-----------------------|
| 1 | Agreement | 22 | 73.3 | 73.3 | 73.3 |
| 2 | Strongly agreement | 6 | 20.0 | 20.0 | 93.3 |
| 3 | No answer | 1 | 3.3 | 3.3 | 96.6 |
| 4 | Disagreement | 1 | 3.3 | 3.3 | 100.0 |
| 5 | Strong disagreement | 0 | 0 | 0 | |
| | Total | 30 | 100.0 | 100.0 | |

VIRTUS 79